Author's response to reviews

Title: Economic analyses to support decisions about HPV vaccination in low and middle income countries: a consensus report and guide for analysts

Authors:

Mark Jit (Mark.Jit@hpa.org.uk)
Carol Levin (clevin@path.org)
Marc Brison (mbrisson@uresp.ulaval.ca)
Ann Levin (annlevin@verizon.net)
Stephen Resch (stephen_resch@harvard.edu)
Johannes Berkhof (H.Berkhof@vumc.nl)
Jane Kim (jim@hsph.harvard.edu)
Raymond Hutubessy (hutubessyr@who.int)

Version: 4 Date: 18 December 2012

Author's response to reviews:

xxx